Author’s response to reviews

Title: Can postponement of an adverse outcome be used to present risk reductions to a lay audience? A population survey

Authors:

- Rasmus Dahl (r.dahl@dadlnet.dk)
- Dorthe Gyrd-Hansen (dgh@dsi.dk)
- Ivar Sonbo Kristiansen (ivarson@c2i.net)
- Jørgen Nexoe (jnexoe@dadlnet.dk)
- Jesper Bo Nielsen (jbnielsen@health.sdu.dk)

Version: 4 Date: 21 March 2007

Author’s response to reviews:

The BioMed Editorial Production Team

Thank you for your letter of 21-03-2007.

We have made the following formatting changes requested:

1. Removed version+date, the text "author" and moved them onto one line, number of tables and word count.

2. Removed "keywords".

3. Removed underlining from the entire document.

4. Cited references using consecutive numbers in square brackets.

5. Checked that the reference list includes all author names.

6. Changed Reference List to References

7. Removed website address from table 1 to References

We hope the paper is now acceptable.

Yours sincerely

Rasmus Dahl
MD